UC Santa Barbara

UC Santa Barbara Previously Published Works

Title

Evidence for a Peak Shift in a Humoral Response to Helminths: Age Profiles of IgE in the Shuar of Ecuador, the Tsimane of Bolivia, and the U.S. NHANES

Permalink https://escholarship.org/uc/item/71n168kd

Journal PLOS Neglected Tropical Diseases, 5(6)

ISSN 1935-2727

Authors

Blackwell, Aaron D Gurven, Michael D Sugiyama, Lawrence S <u>et al.</u>

Publication Date

2011-06-01

DOI

10.1371/journal.pntd.0001218

Peer reviewed

Evidence for a Peak Shift in a Humoral Response to Helminths: Age Profiles of IgE in the Shuar of Ecuador, the Tsimane of Bolivia, and the U.S. NHANES

Aaron D. Blackwell^{1,2,3}*, Michael D. Gurven¹, Lawrence S. Sugiyama^{2,3}, Felicia C. Madimenos^{2,3}, Melissa A. Liebert^{2,3}, Melanie A. Martin¹, Hillard S. Kaplan⁴, J. Josh Snodgrass^{2,3}

1 Integrative Anthropological Sciences, University of California, Santa Barbara, Santa Barbara, California, United States of America, 2 Department of Anthropology, University of Oregon, Eugene, Oregon, United States of America, 3 Institute of Cognitive and Decision Sciences, University of Oregon, Eugene, Oregon, United States of America, 4 Department of Anthropology, University of New Mexico, Albuquerque, New Mexico, United States of America

Abstract

Background: The peak shift model predicts that the age-profile of a pathogen's prevalence depends upon its transmission rate, peaking earlier in populations with higher transmission and declining as partial immunity is acquired. Helminth infections are associated with increased immunoglobulin E (IgE), which may convey partial immunity and influence the peak shift. Although studies have noted peak shifts in helminths, corresponding peak shifts in total IgE have not been investigated, nor has the age-patterning been carefully examined across populations. We test for differences in the age-patterning of IgE between two South American forager-horticulturalist populations and the United States: the Tsimane of Bolivia (n = 832), the Shuar of Ecuador (n = 289), and the U.S. NHANES (n = 8,336). We then examine the relationship between total IgE and helminth prevalences in the Tsimane.

Methodology/Principal Findings: Total IgE levels were assessed in serum and dried blood spots and age-patterns examined with non-linear regression models. Tsimane had the highest IgE (geometric mean = 8,182 IU/ml), followed by Shuar (1,252 IU/ml), and NHANES (52 IU/ml). Consistent with predictions, higher population IgE was associated with steeper increases at early ages and earlier peaks: Tsimane IgE peaked at 7 years, Shuar at 10 years, and NHANES at 17 years. For Tsimane, the age-pattern was compared with fecal helminth prevalences. Overall, 57% had detectable eggs or larva, with hookworm (45.4%) and *Ascaris lumbricoides* (19.9%) the most prevalent. The peak in total IgE occurred around the peak in *A. lumbricoides*, which was associated with higher IgE in children <10, but with lower IgE in adolescents.

Conclusions: The age-patterning suggests a peak shift in total IgE similar to that seen in helminth infections, particularly *A. lumbricoides*. This age-patterning may have implications for understanding the effects of helminths on other health outcomes, such as allergy, growth, and response to childhood vaccination.

Citation: Blackwell AD, Gurven MD, Sugiyama LS, Madimenos FC, Liebert MA, et al. (2011) Evidence for a Peak Shift in a Humoral Response to Helminths: Age Profiles of IgE in the Shuar of Ecuador, the Tsimane of Bolivia, and the U.S. NHANES. PLoS Negl Trop Dis 5(6): e1218. doi:10.1371/journal.pntd.0001218

Editor: Maria Yazdanbakhsh, Leiden University Medical Center, Netherlands

Received December 14, 2010; Accepted May 17, 2011; Published June 28, 2011

Copyright: © 2011 Blackwell et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: Funding came from the National Science Foundation (www.nsf.gov) grants BCS-0824602 and BCS-0422690, the National Institutes of Health (www.nih. gov) R01AG023119-01, UCSB Center for Evolutionary Psychology (NIH 5DP10000516-04 to Leda Cosmides), UO Anthropology Department (http://pages.uoregon. edu/anthro/), Institute of Cognitive and Decision Sciences (http://pages.uoregon.edu/icds/ICDS_ENTER.html), and Office of Research and Faculty Development (http://rfd.uoregon.edu/), and the Ryoichi Sasakawa Young Leaders Fellowship Fund (http://www.sylff.org/). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: ablackwell@isber.ucsb.edu

Introduction

Age-related epidemiological patterns are thought to result from complex interactions between parasite life cycle, exposure to infection, and host-immunity [1,2]. Helminth infections show characteristic age-patterning, peaking around puberty and then declining during adulthood [3–5]. However, this pattern varies with infection prevalence and intensity, tending to both peak and decline earlier in populations with higher rates of transmission. This "peak shift" is thought to result from the interaction between the rate at which new individuals are infected and the rate at which partial immunity is acquired [1,2]. According to this model, when transmission is high infection occurs more quickly, leading to a higher prevalence at a younger age. However, earlier infection also leads to an earlier acquisition of immunity, leading to a decline in prevalence following the peak.

Although studies have found age-patterns in helminth prevalences consistent with this hypothesis, few studies have examined whether the age-patterning of immune responses follows similar patterns. Those studies that have examined the age-patterning of immune responses have generally focused on parasite specific immunoglobulins (IgG, IgA, IgM, and IgE) [6–11]. These have shown age-patterns that resemble the age-specific prevalences of parasites. However, in addition to specific responses, helminth infections are associated with a general shift in the host immune system towards a $T_{\rm H}2$ -biased phenotype, characterized in

Author Summary

Infection with parasitic worms, known as helminths, alters the immune system, causing individuals to produce high levels of a type of antibody known as immunoglobulin E (IgE). IgE is typically very low in western populations, but is many times higher where helminth infections are common, particularly indigenous populations in South America. Helminths infect more than one seventh of the world's population. Since helminths tend to infect people at younger ages in areas where they are more common, a disproportionate number of those affected are schoolchildren. In this paper we examine IgE levels in two indigenous South American groups in comparison to levels in the United States. In these groups we find that IgE levels are not only higher, but that they also reach their highest levels at earlier ages in more infected populations. This finding is important since effects on immune function, including IgE production, may have additional consequences if they occur at young ages, changing the development of allergy and asthma, growth, and response to vaccines.

particular by increased production of total IgE. Although specific responses are thought to participate in protection against infection and thus the generation of peak-shift patterns, an examination of total IgE levels is also critical for understanding the effects of helminths on health and immune function. Due to stimulation of T_H2 responses, total IgE is likely to represent the total burden of multiple helminth species infection better than species specific immunoglobulins. Like specific IgE, total IgE levels are elevated in infected individuals and fall with treatment [12,13], and have been shown to correlate with specific IgE for *Ascaris lumbricoides* and *T. trichiura* [14]. However, total IgE levels in heavily parasitized individuals remain elevated compared to individuals in industrialized countries for substantial periods of time [15], suggesting persistent changes in host immune function.

Mounting an immune response is energetically costly, necessitating reductions in competing life history demands, including growth, reproduction, and survival [16–17]. Higher total IgE levels are associated with poorer growth and shorter adult stature, suggesting a trade-off between growth and investment into immune response [18]. Moreover, the shifting of immune function towards a $T_{\rm H2}$ phenotype may reduce $T_{\rm H1}$ responses, decreasing the effectiveness of vaccines or increasing susceptibility to viruses and bacteria [19–21].

These effects may depend, in part, on the timing of exposure, as exposure to helminths during critical periods may bias the development of immune function or a child's growth trajectory. Helminths infect more than one seventh of the world's population, and given the peak-shift pattern, a disproportionate number of those infected are schoolchildren [22]. As a consequence, agepatterns in helminth infection and immune response are likely to have significant consequences on growth and development.

Although several studies have reported that IgE increases quickly in the first 5–10 years of life and then levels off [23,24], few studies have carefully examined the age-patterning of total IgE and we know of no published studies that have compared agepatterning in IgE across multiple populations. As a marker of helminth infection and T_H 2-biasing of T-cell responses, an understanding of the age-patterning of total IgE is important for understanding the broader consequences of helminth infections on life history parameters. The current study describes in detail the age-patterning of IgE levels in three populations. These include data from the United States collected by the National Health and Nutrition Examination Survey 2005–2006 (NHANES) and data from two populations of South American forager-horticulturalists: the Tsimane of Boliva and the Shuar of Ecuador. First, we test for predicted associations between population mean IgE level and the age-pattern of IgE. Second, using Tsimane data we examine the relationship between age-patterning in IgE and age-patterning in helminth infections.

Methods

Study Populations

Shuar. Shuar are Amerindians from the Amazonas region of Ecuador [25,26]. Shuar live across a wide range of circumstances, but a large portion of the population continues traditional subsistence based on horticulture, hunting, and fishing. Approximately 40% of Shuar children are stunted, a higher prevalence than is found in other indigenous and non-indigenous children living in the same area [27]. Although we know of no studies examining helminth infections in the Shuar, recent studies report infection rates of around 50% in other Amazonian Ecuadorian populations, with *Ascaris* the most prevalent parasite [28,29]. Shuar data were collected as part of the Shuar Life History Project (www.bonesandbehavior.org/shuar) in a village that has been previously described [18].

Tsimane. Tsimane are forager-horticulturalists that live along the Maniqui River in lowland Bolivia. Tsimane subsist primarily on cultivation of plantains, rice, manioc, and corn, as well as hunting, fishing, and gathering. Tsimane show high levels of inflammatory markers, such as C-reactive protein [30–32]. Helminth infections are highly prevalent, with hookworm (*Necator americanus* or *Ancylostoma duodenale*) being the predominant parasite, infecting between 44% and 76% of children [33,34]. Between 40– 50% of children are stunted [35,36]. The data for this study were collected as part of the Tsimane Health and Life History Project (http://www.unm.edu/~tsimane/), in sixteen villages representing a range of environmental and economic situations (interior forest, riverine, acculturated, non-acculturated).

National Health and Nutrition Examination Survey (NHANES). NHANES is a large-scale, national survey of health, nutrition, and social factors conducted by the National Center for Health Statistics and Center for Disease Control. This study uses data from the NHANES 2005–2006 dataset (http://www.cdc.gov/nchs/nhanes/nhanes2005-2006/nhanes05_06.htm). The sample includes 8,336 individuals, 88% percent U.S. citizens, 52% percent female, 27% Mexican-American, 36% Caucasian, 26% African-American, and 11% other ethnicities.

Ethics Statement

For Shuar, permission to conduct the study was first obtained from the Federacion Interprovincial de Centros Shuar (FICSH), the elected representational organization for Shuar affairs. Second, permission was obtained from elected village leaders. Third, a village meeting was held in which a village-level consent form was read aloud, the study explained, questions answered, and a community decision reached about whether to allow the study. Individuals were informed that they could choose not to participate, participate only in individual portions of the study, or participate in the full study. At the time of data collection, individual oral consent was obtained, with individuals able to optin or out of individual components of the study (e.g., to provide blood spots or not). For subjects under age fifteen (the local age of consent) both parental consent and child assent were obtained. Oral consent was used for two reasons: 1) many Shuar are non or semi-literate or have only a few years of schooling, and 2) many Shuar are suspicious or uncomfortable with signing documents due to a history of territorial land disputes and wariness about signed documents leading to ownership conflicts. An independent bilingual Shuar village leader, nurse, FISCH official or assistant was present to translate as needed during group and individual consent and study procedures. The study and consent procedures were approved by the Institutional Review Board (IRB) of the University of Oregon.

For Tsimane, informed consent was obtained at three levels: 1) from the Gran Consejo Tsimane, the local Tsimane government organization that represents Tsimane interests and oversees all projects, 2) community officials and participants in village meetings, and 3) individual consent during medical visits and before each procedure. After explanation of a formal protocol by bilingual Tsimane assistants, consent forms were signed for literate participants, and verbal approval with fingerprint signature given for non-literate participants. Tsimane consent procedures were approved by the IRBs at the University of New Mexico, University of California, Santa Barbara and the University of Southern California.

Blood Collection and Analysis

Shuar. Shuar samples were collected following standard procedures to collect dried blood spots [37]. IgE levels were determined by ELISA at the University of Oregon, following a commercially available protocol (Bethyl Labs, Inc.: #E80-108 and #E101) adapted for use with blood spots [38]. Blood spot collection and IgE analyses have been previously described [18].

Tsimane. Tsimane blood samples were collected by venipuncture during annual medical exams conducted by Bolivian physicians. Serum was frozen in liquid nitrogen for transport to New Mexico. Two rounds of samples were collected. The first 223 samples were collected in 2004-2005 and analyzed by TriCore Laboratories (Albuquerque, NM) for total IgE (catalog: L2KIE6) using an Immulite 2000 (Siemens Corp; Deerfield, IL). An additional 700 samples were collected in 2007 and analyzed in the laboratory of JJS at the University of Oregon using the same commercial ELISA kit used for Shuar samples (Bethyl Labs, Inc.: #E80-108 and #E101). Of these, 91 were repeated measures for individuals included in the first batch of samples. These samples were excluded so as not to confound longitudinal and cross-sectional data. After excluding these cases, the first and second samples did not differ in geometric mean IgE (comparison of log transformed IgE: t = .462, df = 830, p = .644).

NHANES. NHANES samples were collected by venipucture by trained phlebotomists. Determination of total IgE was done using the ImmunoCAP 1000 system (Pharmacia Diagnostics) by the Department of Pathology Immunology Laboratory at Elmhurst Memorial Hospital, Elmhurst, IL. Details can be found at http://www.cdc.gov/nchs/data/nhanes/nhanes_05_06/al_ige_d_met_specific_ige_total_ige.pdf.

Comparability of Blood Samples. A handful of studies have compared IgE in dried blood spots to IgE measured in serum and found results to be virtually identical [39,40]. Additionally, the ELISA procedure used to determine blood spot IgE in this study has been validated against controls with known IgE levels [38]. To verify the comparability of IgE obtained from dried blood spots with that obtained from serum, six matched blood spot and serum samples were analyzed using both methods. The values obtained from dried blood spots were highly correlated with the values from serum (r = 0.98, p < .001). Blood spot values were $\sim 3\%$ higher. Using linear regression, the following conversion factor was

obtained to convert dried blood spot values into serum values prior to other analyses: $IgE_{serum} = 0.965 \times IgE_{DBS} - 3.458$ (IU/ml).

Fecal Analysis

Tsimane fecal samples were analyzed using two methods. From 2004 to 2008 fecal samples were analyzed for the presence of helminth eggs and larvae by direct identification on wet mounts. As described in greater detail elsewhere [33], duplicate mounts were prepared with 0.9% saline solution and iodine solution, respectively, and examined at 100x and 400x for helminth eggs (hookworm, *A. lumbricoides*, and *T. trichiuris*), and larvae (*S. stercoralis*). Beginning in 2007, fecal samples were also preserved in 10% formalin solution following direct identification, and later quantitatively analyzed using a modified Percoll (Amersham Pharmacia) technique [41].

Of the two methods, the Percoll technique is more sensitive, producing slightly higher detection rates than direct identification (59.4% vs. 51.9% infected). These differences may be due to the greater efficiency of the Percoll technique in detecting eggs in fibrous stools and at low-intensities [41]. However, for the present study the differences between the two methods were not qualitatively great enough to justify using only data produced by one method or the other. We therefore aggregated data from the two methods, coding individuals as either infected or not infected if helminths were detected by either method. In total 1,495 individuals had Percoll results, with the remaining 3,610 having only direct results.

Age Estimation

Birth dates accurate to the month were available from health clinic and school records for most Shuar children. For Shuar adults, birth dates on government identification were cross-checked with extensive genealogical information collected from multiple informants. Tsimane genealogies were collected during demographic interviews done on individuals over age 18 (n = 1,098). Tsimane ages were estimated based on written records, such as those kept by Catholic missionaries, demographic interviews with independent cross-checking of genealogies and reproductive histories with multiple informants, and the use of photographs of people with known ages [42].



Figure 1. Distribution and geometric mean value of IgE antibody levels in three populations. Density plots were generated with a Gaussian smooth with bandwidth 0.5. doi:10.1371/journal.pntd.0001218.g001

Data Analysis

Prior to data analysis, IgE values were converted into international units (1 IU = 2.4 ng/ml). IgE is log-normally distributed in all three populations (Figure 1), so values were natural log transformed (lnIgE) before all analyses. For t-tests, reported means are geometric means calculated by taking the exponential of the mean log values used in the t-test. Descriptive statistics and t-tests were done in PASW Statistics 18.0 (formerly SPSS Statistics, SPSS Inc.). All other analyses were done in R 2.10.1 (www.r-project.org).

Generalized additive models (GAM; [43,44]) were used to examine the non-parametric age pattern of IgE levels for each population. Models were fit with the *gam* procedure in package *mgcv* using thin plate regression splines [45,46]. Since the cases in each population were not evenly distributed by age, initial basis knots were specified for each population based on even tenpercentiles of the age distribution, allowing knots to be spaced with an equal number of cases between them (Figure 2). Apart from the basis knots, smoothing parameters were generated automatically according to *gam* defaults [45]. GAM models included an intercept, a sex factor, a spline for age, and a spline for age-bysex interaction.

In initial models, similar IgE levels at birth were predicted among Shuar and NHANES, with the Shuar model predicting IgE of 7 IU/ml for females and 9 IU/ml for males, and the NHANES model predicting 15 IU/ml for females and 21 IU/ml for males. However, due to the relatively low number of Tsimane under age five, initial Tsimane models were essentially straight lines, with peak IgE predicted at birth. A number of studies have found



Figure 2. Models for IgE by age in Tsimane, Shuar, and NHANES. A) Generalized additive models for Tsimane (top, blue), Shuar (middle, green), and NHANES (bottom, yellow). Points show the mean InIgE value for males (triangles) and females (circles) between knots specified in the initial model basis (vertical lines), while lines indicate the thin plate regression spline for each sex. For all three populations males have the higher fit line. Numbers indicate the estimated ages at which the initial peak in IgE occurs. Shading indicates local 95% confidence intervals for the spline, with dark areas indicating overlap between male and female confidence intervals and light areas indicating no overlap. B) Ordinal step models and non-linear regression models. Ordinal model parameters were entered in stepwise fashion according to AIC minimization, resulting in the final models. Numbers indicate the significant transition at greater than the age given, symbols the significance of the parameter in the model: $t p \leq 0.01$, ** $p \leq 0.01$. Dashed lines indicate the model fits for non-linear models, including the population specific models in Table 3 (red), and the interaction models in Table 4 (Model 1 in green, Model 2 in brown, and Model 3 in blue). For simplicity only the models for females are shown. doi:10.1371/journal.pntd.0001218.g002

. www.plosntds.org

extremely low IgE levels at birth (<1 IU/ml) [47-50,23,51,52], even in infants of mothers with helminth infections and high IgE [53,54]. Given the convergence of the other two models and these previous findings, we used dummy cases with age zero and IgE equal to 15 IU/ml to anchor Tsimane models to a similar intercept at birth. Dummy cases were included in GAM models but not in any other statistic.

GAM models with a binomial logit-link function were also used to estimate odds-ratios for Tsimane helminth infection by age. Associations between helminth infection and IgE levels were estimated in linear models controlling for infection with other helminths, sex, and age.

In addition to GAM, two other methods were used to verify age shapes and compare populations. In the first, a stepwise linear regression was used to identify critical age-related changes in InIgE for each population. Dummy variables were coded for each unique age indicating whether a case was greater than the given age (e.g., [55]). Starting from a model with only an intercept and sex term, *stepAIC* (package MASS) was used to enter and remove age variables to minimize model AIC [56].

For the second test we constructed non-linear models composed of linear segments linked together, with model terms representing the point at which the linear segments are stitched together. In this model, model terms directly represent critical ages, such as the age at which the model peaks, so differences in critical ages between populations can be tested using population interaction terms. The basic model is:

$$ln(IgE) = \beta_0 + \beta_S Sex + \sum_{i=1}^{3} \left(\beta_i \left[\frac{Age - \alpha_{i-1}}{1 + e^{-10(Age - \alpha_{i-1})}} \right] \times \left(1 - \frac{1}{1 + e^{-10 \times (Age - \alpha_i)}} \right) + \frac{\alpha_i - \alpha_{i-1}}{1 + e^{-10(Age - \alpha_i)}} \right) + \varepsilon$$

Where α_1 , α_2 and α_3 are the ages at which the slope changes and α_0 equals zero, β_1 , β_2 , and β_3 are the slopes for the segments, and additional terms β_0 and β_s represent the intercept at age zero and sex effect, respectively. Three logistic functions serve to "turn-on",

"turn-off", and maintain the value reached at each age transition. The models were fit such that α_1 is the initial age where rapid increases in IgE level out, α_2 is the age at which IgE peaks for the population, and α_3 is the age at which IgE reaches mean adult levels. Models were solved using *nls* (package stats) using the nl2sol algorithm. Initial values specified based on GAM regressions and only individuals under age 50 were used for modeling.

Results

Age-Pattern of IgE

Of the three populations, Tsimane had the highest IgE levels (geometric mean =8,182 IU/ml), followed by Shuar (1,252 IU/ml), and NHANES (52 IU/ml) (Table 1). IgE distributions were skewed but largely normalized by log-transformation (Figure 1). All three groups differed from one another in pair-wise comparisons (all t-test p-values <.001 after Bonferroni correction). In all three populations, males had higher IgE than females. NHANES males had IgE levels 60% higher than females (65.9 vs. 41.3 IU/ml, t=14.08, p<.01), while Shuar males had IgE values 29% higher than Shuar females (1,457 vs. 1,129 IU/ml, t=2.15, p=.03), and Tsimane males had levels 16% above females (8,720 vs. 7,527 IU/ml, t=2.73, p<.01).

Upon initial visual examination of the data, age patterns were observed to be non-linear. We therefore used thin plate regression splines in GAM models to examine the age patterning of IgE (Figure 2A). Age terms were significant in all models (Tsimane: edf = 10.94, F = 43.15, p < .001; Shuar: edf = 7.73, F = 5.35, p<.001; NHANES: edf=8.93, F=24.74, p<.001). Despite differences in level, all three populations had similar age-related IgE profiles, characterized by a rapid increase before age five, a peak in the juvenile or adolescent period, and a decrease into adulthood. However, a number of features differ between populations. Principal among these is the age at which IgE initially peaks. Tsimane IgE peaked at 7.3 years for males and 7.2 years for females. Shuar IgE peaked at 10.2 for both sexes. NHANES IgE did not peak until age 16.9 for males and 16.4 for females. Fitting a linear model to the three population points for each sex suggested that for males the peak age decreases by 1.98

Table 1	 Sample 	sizes and	lgE by	population	and age	category.
---------	----------------------------	-----------	--------	------------	---------	-----------

	Tsima	Tsimane			Shuar			NHANES		
Ages	N	Geometric Mean		N	Geometr	Geometric Mean		Geom	Geometric Mean	
1–5	11	7,073	(1,078–46,426)	50	1,017	(127–8,124)	938	34	(2–706)	
6–10	79	9,755	(1,989–47,847)	102	1,658	(308–8,920)	774	61	(3–1,288)	
11–15	51	7,798	(1,788–34,008)	35	1,307	(172–9,927)	1,110	64	(3–1,380)	
16–20	38	7,820	(2,411–25,364)	12	1,055	(109–10,228)	1,110	69	(3–1,453)	
21–30	82	8,068	(1,641–39,670)	22	882	(64–12,205)	931	53	(2–1,131)	
31–40	90	7,887	(1,594–39,031)	36	1,074	(178–6,491)	778	47	(3–882)	
41–50	219	8,321	(1,797–38,534)	17	1,024	(167–6,289)	763	50	(2–1,058)	
51–60	119	7,861	(1,385–44,612)	5	445	(78–2,547)	598	49	(3–763)	
61–70	82	8,018	(1,534–41,916)	5	1,738	(182–16,623)	615	54	(2–1,319)	
71–80	48	9,037	(2,590–31,537)	3	2,327	(651–8,311)	427	39	(2–863)	
81–90	12	5,068	(620–41,428)	2	1,767	(160–19,485)	292	41	(2–1,028)	
Total	831	8,182	(1,691–39,582)	289	1,252	(172–9,118)	8,336	52	(2–1,116)	

Geometric mean values are in IU/ml. Values in parenthesis are plus or minus two standard deviations for log transformed values: $e^{\mu \pm 2^*\sigma}$, where μ is the mean of ln(lgE) and σ is the standard deviation of ln(lgE).

doi:10.1371/journal.pntd.0001218.t001

1

years for every one unit increase in population mean lnIgE (t = 18.40, p = 0.04), while for females the peak age decreases by 1.76 years per unit increase in lnIgE (t = 25.25, p = 0.03).

We next used stepwise linear regression with ordinal age variables to identify ages at which important transitions in IgE level occur and to test the significance of these changes (Figure 2B). Tsimane transitions included an increase at age three ($\beta = 1.06$, t = 1.90, p = 0.06) followed by a decrease after age nine ($\beta = -0.21$, t = -2.26, p = 0.02). For Shuar, there were significant increases after age two ($\beta = 1.07$, t = 2.10, p = 0.04) and age three ($\beta = 1.18$, t = 3.75, p<0.01), and a significant decrease after age eleven ($\beta = -0.41$, t = -3.59, p<0.01). In the NHANES sample increases were present in the model after age one ($\beta = 0.48$, t = 3.82, p<0.01), age three ($\beta = 0.22$, t = 1.60, p = 0.10), age four ($\beta = 0.27$, t = 1.90, p = 0.05), age six ($\beta = 0.15$, t = 1.67, p = 0.10), and age fifteen ($\beta = -0.13$, t = 2.00, p = 0.04), with a decrease after age eighteen ($\beta = -0.19$, t = -2.79, p<0.01).

Since neither of these models directly tests for differences between populations or allows for the comparison of shape differences in age curves or peaks, we devised a non-linear modeling procedure in which four linear segments are used to model the age profile (Figure 2B). These models include three ages points that correspond to the point at which the rapid increases in early life levels off (α_1), the age at which IgE peaks in the population (α_2), and the age at which IgE reaches adult levels after the peak (α_3). Three slopes (β_{1-3}) describe the change in IgE between age points (birth – α_1 , α_1 to α_2 , and α_2 to α_3). A sex term accounts for the difference between males and females (β_s).

Models were first fit for the three populations independently (Table 2). Model parameters conformed well to predictions from GAM models, with peak ages (α_2) of 8.2, 10.0, and 17.9 predicted for Tsimane, Shuar, and NHANES respectively. The ages of

Evidence for a Peak Shift in IgE

initial slope change and final adult level also corresponded to peak ages, with both ages earliest in Tsimane and latest in NHANES. Moreover, all model parameters were highly correlated with IgE levels (Figure 3). Age terms, initial slopes from age zero, and sex differences all correlated with mean log IgE (α_1 : r = -1.00,p<0.01; α_2 : r = -0.98, p = 0.13; α_3 : r = -1.00,p = 0.01; β_1 : r = 0.99, p = 0.02; β_s : r = -0.99, p = 0.09), while the increase between the first peak and the final peak, and the decrease from the final peak to adult levels correlated with untransformed population geometric mean IgE (β_2 : 1.00, p = 0.06, β_3 : -1.00, p<0.01).

To compare populations on these terms we first attempted to fit models with population interaction terms for each parameter. However, this model, with 21 parameters, was too complex for the model algorithms and the data available, and failed to fit. Instead we simplified the models based on the relationship between model parameters and population mean IgE levels. In the first of these models we included parameter by population IgE interaction terms (Table 3, Model 1). This model verified interactions between population IgE and all model parameters, with each one unit increase in log IgE associated with a 0.37 year decrease in the age of the initial slope change, a 1.70 year decrease in the age of peak IgE, and a 4.10 year decrease in the age at which levels dropped to adult mean values. The initial rate of increase in IgE was also significantly related to IgE mean levels indicating bother faster and earlier acquisition of high IgE in the Tsimane and secondarily the Shuar.

In the second, third, and forth models we tested population differences in the ages at which slopes change, using population factor terms rather than interactions with population IgE. In Model 2 all three ages were left independent and the Shuar were used as a contrast group, since they lie between Tsimane and NHANES. In this model NHANES α_1 and α_2 were significantly

Population	Parameter	Estimate	SE	t-value	р
NHANES	β_1 (Initial Slope)	.21	.02	13.57	<.001
	β_2 (Second Slope)	1.43×10 ⁻²	9.09×10 ⁻³	1.58	.115
	β_3 (Post-peak Decline)	-4.24×10^{-2}	1.83×10 ⁻²	-2.32	.021
	α_1 (Age of slope change)	5.33	.52	10.21	<.001
	α_2 (Age of peak)	17.85	1.56	11.41	<.001
	α_3 (Age adult level reached)	25.55	2.52	10.13	<.001
	β_{s} (Male vs. Female)	.44	.04	11.95	<.001
Shuar	β_1 (Initial Slope)	1.15	.09	12.62	<.001
	β_2 (Second Slope)	2.73×10 ⁻²	5.04×10 ⁻²	.54	.588
	β_3 (Post-peak Decline)	-8.25×10^{-2}	8.64×10 ⁻²	96	.340
	α_1 (Age of slope change)	3.91	.36	10.94	<.001
	α_2 (Age of peak)	9.99	3.13	3.19	.002
	α_3 (Age adult level reached)	15.94	4.29	3.72	<.001
	β_{s} (Male vs. Female)	.22	.11	2.02	.044
Tsimane	β_1 (Initial Slope)	1.78	.23	7.86	<.001
	β_2 (Second Slope)	.26	.09	2.86	.004
	β_3 (Post-peak Decline)	33	2.20	15	.882
	α_1 (Age of slope change)	3.06	.53	5.82	<.001
	α_2 (Age of peak)	8.21	2.85	2.88	.004
	α_3 (Age adult level reached)	9.92	6.18	1.61	.109
	β_{S} (Male vs. Female)	.15	.06	2.30	.022

Table 2. Non-linear model parameters by population.

doi:10.1371/journal.pntd.0001218.t002



Figure 3. Association between model parameters and population geometric mean IgE levels. The upper panel shows fits between age parameters and mean log IgE by population. The lower shows fits between slope parameters and geometric mean IgE by population. Note that the fit for B1 is linear with regard to log IgE, but is shown on the lower graph due to the parameter scale. Correlation coefficients for all parameters are given in the text. doi:10.1371/journal.pntd.0001218.g003

later than Shuar ages, and α_3 was later but with marginal significance. Although all three Tsimane ages were early than Shuar ages, none were significantly so, although all three were significantly earlier than NHANES ages when the model was run with NHANES as the contrast group (not-shown). Given the strong correlation between all three ages and mean IgE levels, we suspected that multicollinearity between terms might be reducing parameter significance. We therefore examined how α_1 , α_2 , and α_3 might be included as functions of a single age term. By fitting linear models to the parameters in Table 3 we found that $\alpha_1 \sim 1.47 + \alpha_2 \times 0.22$, and $\alpha_3 \sim 0.90 + \alpha_2 \times 1.50$. We used these terms in Model 3, removing α_1 and α_3 . In this model with a single age term to describe the shape, ages in both Tsimane and NHANES were significantly different from Shuar ages, with the overall age shape shifted earlier in Tsimane and later in NHANES.

Tsimane Age-Pattern of Helminth Infection

Overall, 57% of Tsimane participants were infected with at least one helminth species, with hookworm (45.3%) and *A. lumbricoides* (19.88%) the most prevalent, and *S. stercoralis* (5.6%) and *T. trichiura* (3.2%) less common (Table 4). In order to compare the IgE agepattern with helminth infection patterns, we examined likelihood of helminth infection by age in the Tsimane sample using logistic GAM models (Figure 4). By sex the only significant difference was in A. lumbricoides infection, with women being more likely to be infected (22% vs. 18%, $\chi^2 = 15.5$, p<.001). By age, the odds-ratio of hookworm infection is highest in adults over age 45, but also has a small peak at age 12.8. In contrast, the odds-ratio for infection with A. lumbricoides peaks sharply at age 8.1 and then declines, mirroring the IgE age-pattern more closely. The odds-ratio for infection with S. stercoralis peaks somewhat later, around age 24.9. The odds of T. trichiura infection is essentially flat with respect to age, reflecting the low prevalence of T. trichiura. Overall, the oddsratio for having any type of helminth infection peaks at age 11.1 and then declines until age 45, at which point it increases again. Odds-ratios closely mirror actual prevalences by age group (Table 4). For infected individuals we also examined whether egg/larva burden showed age-patterning. The only significant age-pattern was a slight decline in hookworm burden with age up to about age thirteen (not shown). Other egg/larva burdens did not show age-patterning independent of changes in detection prevalence.

Association between Helminth Infection and IgE Levels

We examined the association between helminth infection and IgE levels in our Tsimane sample using regression models to control for co-infection status, and with the sample divided by age group (Figure 5). Hookworm infection was significantly associated with higher IgE levels in 11–20 year-olds (β =0.44, t=2.85, p<0.01), individuals over forty (β =0.25, t=3.35, p<0.01), and in the overall sample (β =0.23, t=4.08, p<0.01). *A. lumbricoides* infection was significantly associated with higher IgE levels in 11–20 year-olds (β =-0.45, t=-2.45, p=0.02). Although non-significant, *T. trichiura* infection showed a pattern similar to *A. lumbricoides* in those 10 and younger (β =0.84, t=1.41, p=0.16). *S. stercoralis* was positively associated with IgE levels only considering the overall sample (β =0.32, t=2.24, p=0.03).

From the total sample, 459 individuals had IgE levels and full Percoll egg/larva counts. Of these, 195 were positive for hookworm, 89 for *A. lumbricoides*, 18 for *S. stercoralis*, and 16 for *T. trichiura*. Examining infected individuals only, egg/larva counts were not significantly correlated with lnIgE, either in the overall sample or with the sample dived by age.

Discussion

We report on the age patterning of IgE in three populations: U.S. residents, Ecuadorian Shuar, and Bolivian Tsimane. The highest known IgE levels are found among lowland South American populations [18]. Tsimane IgE levels fit this pattern and resemble the levels of other South American groups with low levels of market integration (e.g., [14,57,58]). Tsimane levels are significantly higher than typical values in the United States, even for individuals reporting high levels of allergic symptoms (based on NHANES data, analysis not shown). In contrast, despite inhabiting a similar neotropical environment, Shuar display lower IgE, resembling other South Americans living in rural areas [12,13,59]. However, Shuar IgE was also significantly higher than NHANES values.

Although many studies have reported elevated IgE levels in populations infected with parasites such as helminths and malaria, very few have carefully characterized the age-patterning of IgE, Table 3. Four models comparing age profile parameters among NHANES, Shuar, and Tsimane.

	Unit	Model 1		Model 2		Model 3	
Parameter		Estimate	SE	Estimate	SE	Estimate	SE
β_1 -intercept	10 ⁻¹ x IU/ml/yr	-9.22****	1.16	-9.82***	1.50	-10.13***	0.96
β_1 -slope	10 ⁻¹ x IU/ml/yr/ In(IgE)	2.89****	0.30	3.03***	0.37	3.11***	0.24
β_2 -intercept	10 ⁻² x IU/ml/yr	1.44	0.93	1.26	0.88	1.29 ^t	0.79
β_2 -slope	10 ⁻⁵ x IU/ml/yr/Mean IgE	3.50 ^t	1.83	2.96	2.08	2.52	2.05
β_3 -intercept	10 ⁻² x IU/ml/yr	-2.19**	0.82	-3.01*	1.48	-4.08****	0.78
β_3 -slope	10^{-5} x IU/ml/yr/ Mean IgE	-8.47^{t}	4.80	-5.20	10.33	-1.44	1.15
α ₁ -intercept	Years	6.55****	.92				
α ₁ -slope	Years/Mean In(IgE)	-0.37*	.15				
α ₂ -intercept	Years	23.70***	3.48				
α ₂ -slope	Years/Mean In(IgE)	-1.70***	.39				
α ₃ -intercept	Years	46.12***	5.94				
α ₃ -slope	Year/Mean In(IgE)	-4.10***	.66				
β _s (Male)	10 ⁻¹ x IU/ml	6.77***	1.09	6.84***	1.10	6.77***	1.10
β _{S-slope}	10 ⁻¹ x IU/ml/Mean In(IgE)	-0.59^{*}	.23	-0.61**	.24	-0.59^{*}	.23
α ₁ (Shuar)	Years			3.85***	.42		
α_1 (NHANES vs. Shuar)	Years			1.49*	.64		
α_1 (Tsimane vs. Shuar)	Years			-0.64 ^t	.35		
α ₂ (Shuar)	Years			9.85***	2.34	10.09****	.94
α_2 (NHANES vs. Shuar)	Years			7.66**	2.89	7.66***	1.27
α_2 (Tsimane vs. Shuar)	Years			-1.47	1.98	-2.08*	1.04
α ₃ (Shuar)	Years			15.92**	5.96		
α_3 (NHANES vs. Shuar)	Years			11.63 ^t	6.65		
α_3 (Tsimane vs. Shuar)	Years			-6.21	4.89		

All models are non-linear regression models of the form given in the Methods section of this paper. In Model 1 all parameters were entered as functions of population IgE level. For Models 2 and 3 age parameters were instead entered with population identity interaction terms. In Model 3 α_1 and α_3 were entered as functions of α_2 . Two-sided t-test significant levels:

^tp≤0.10.

*p≤0.05.

doi:10.1371/journal.pntd.0001218.t003

and none that we are aware of has tested for a peak shift. A number of studies have noted that IgE is very low at birth, but increases rapidly in the first five years of life [23,24,52,60]. However, most of these studies have been conducted in North

America or Europe, and most report that IgE is relatively stable after age five or six without characterizing the degree of stability. One of the few studies to report detailed age profiles found an initial increase to age nine, a slight decrease, and then a second

Table 4. Tsimane helminth prevalences by age group.

Age Group	n	Hookworm	Ascaris Iumbricoides	Strongyloides stercoralis	Trichuris trichiura
0–9	1715	41.3%	22.4%	5.9%	3.1%
10–20	838	48.2%	20.9%	6.7%	3.7%
20–30	598	42.5%	20.1%	8.4%	3.3%
30–40	577	44.4%	20.3%	4.0%	3.3%
40–49	607	49.6%	16.8%	3.6%	3.3%
50–59	342	54.1%	16.4%	4.4%	2.9%
60+	375	53.1%	16.3%	4.3%	2.4%
Total	5,105	45.4%	1 9.9 %	5.6%	3.2%

doi:10.1371/journal.pntd.0001218.t004

^{**}p≤0.01. ***p≤0.001.



Figure 4. Odds-ratios for Tsimane helminth infection by age relative to the Tsimane population as a whole. Odds-ratios were estimated with generalized additive models with a binomial logit link function. Shading shows the 95% confidence interval for the odds-ratio. All age functions were significant at p<0.001, except the age function for *T. trichiura*, which was non-significant. doi:10.1371/journal.pntd.0001218.g004

www.plosntds.org





peak at age fifteen in Croatian children [61]. The shape of the increase, with an initial peak and then a second peak, is remarkably similar to the age profiles seen in this study for the NHANES and Shuar sample. Our results suggest that IgE does reach an initial plateau between ages three and five, but continues to increase slowly before reaching higher peaks at age seventeen in the U.S, age ten in the Shuar, and age seven in the Tsimane. It is important to note that although we report the peaks for simplicity, the overall shape of the pattern is more important than the peak itself. This includes a faster rate of increase at an earlier age, an earlier peak, and an earlier decline to adult levels.

The age-patterns we report in this study are consistent with mathematical models for what is known as the peak shift [1,2,62]. The peak shift model predicts that immunity will develop earlier in populations with higher exposure and transmission rates and subsequently decline earlier as cohorts acquire partial immunity. The peak shift hypothesis was formulated with regard to helminth infections. Typically, helminth infections peak just before or during adolescence [1,4,5,22]. Although data on helminth infections was only available for one of our three populations, we suggest that the IgE levels and peak ages reported in this study for Shuar and Tsimane are likely the consequence of high helminths loads since helminths are a primary cause of elevated IgE in rural populations. The IgE patterns reported also match expectations from helminth infections. The lack of helminth data for the NHANES participants may also not be much of a limitation, as what studies exist support the assumption that helminths among US residents are likely to be much less prevalent than among either Shuar or Tsimane. There are few recent estimates, but in 1972 Warren estimated that 4.0 million Americans were infected with A. lumbricoides, 2.2 million with T. trichiuris, 0.7 million with hookworm, and 0.4 million with S. stercoralis [63]. Given the US population in 1972, these are prevalences of 1.9%, 1.0%, 0.3%, and 0.2% respectively. Hotez revises Warren's estimate for *S. stercoralis* to a current estimate of 68-100,000 or 0.05% of the 2008 population [64]. Similarly, of 216,275 stool samples sent to state laboratories in 1987, only 0.8% were positive for *A. lumbricoides*, 1.2% for *T. trichiuris*, 1.5% for hookworm, and 0.4% for *S. stercoralis* [65]. A similar study examined 2,896 samples sent to state laboratories in 2000 and found that 0.4% were positive for *A. lumbricoides* [66]. These estimates are clearly much lower than the prevalences we report for Tsimane and the prevalences reported for other ethnic groups living near the Shuar, enough so that the exact prevalence is not critical for interpreting our findings.

Due to T_{H2} biasing, total IgE may be a better index of total helminth load than specific IgE levels. However, the lack of parasite-specific IgE in these data sets is also a limitation in that we cannot state how much parasite-specific IgE contributes to total levels. It may be that Shuar and Tsimane differ less in the total helminth prevalences than they do in prevalences of particular helminth species. Using helminth infection data for the Tsimane we were able to examine associations between helminth species and total IgE. We found that the overall age-pattern for IgE in the Tsimane resembled the age-pattern for A. lumbricoides infection. A. lumbricoides infection was associated with higher IgE levels in children age 3-10, but with lower IgE levels in 11-20 year-olds. These data suggest that this species may contribute more to the age-pattern of IgE than others. Hookworm and Strongyloides infection were also associated with higher IgE in the overall sample, but showed less age-pattern in association. Future studies will need to investigate this in more detail by examining specific-IgE and extending into other populations.

The association between IgE and *A. lumbricoides* is consistent with other studies showing that total IgE is correlated with specific IgE to *A. lumbricoides* [14]. The positive association between *A. lumbricoides* and IgE in participants under age ten and the negative association after age ten may also suggest that IgE conveys partial immunity to *A. lumbricoides* [67,68]. Other studies that [69] have failed to find increased resistance with higher IgE may have not taken this age-pattern into account.

Other parasites, such as *Plasmodium falciparum*, also raise total IgE levels [70]. However, malaria is unlikely to be an important factor for the populations studied in this paper. Although malaria is present in parts of Shuar territory, it is not present in the villages where the data for this paper were collected, and very few individuals in the area report having had it. Malaria also appears to be absent from the Tsimane territories, with no Tsimane reporting malaria in extensive health interviews.

Finally, in all three populations IgE levels were higher in males. Although noted in many studies (e.g., [23]), the reason for this sex difference is not entirely clear. The only significant sex difference in helminth infections was in *A. lumbricoides*, with slightly more women being infected. Due to the higher IgE in males, it is tempting to hypothesize that this is due to increased resistance in males. However at present this is merely supposition. It is just as likely that Tsimane women are infected more frequently because they spend more time in direct contact with children, who themselves have the greatest number of *A. lumbricoides* infections.

In addition to its importance for theoretical models describing the epidemiology of infections, an understanding of the age patterning of IgE may have public health implications. In populations with higher parasite transmission rates, exposure triggers an elevation of IgE at earlier ages. More rapid and heavy investment in earlier immunocompetence may be favored with high exposure, even at the expense of other investments, such as

References

- Woolhouse M (1998) Patterns in Parasite Epidemiology: The Peak Shift. Parasitol Today 14: 428–434. doi:10.1016/S0169-4758(98)01318-0.
- Anderson RM, May RM (1985) Herd immunity to helminth infection and implications for parasite control. Nature 315: 493–496. doi:10.1038/315493a0.
- Wahyuni S, Sartono E, Supali T, Zee JS van der, Mangali A, et al. (2005) Clustering of allergic outcomes within families and households in areas endemic for helminth infections. Int Arch Allergy and Immunol 136: 356–364.
- Faulkner H, Turner J, Kamgno J, Pion SD, Boussinesq M, et al. (2002) Age-and infection intensity-dependent cytokine and antibody production in human trichuriasis: the importance of IgE. J Infect Dis 185: 665–672. doi:10.1086/ 339005.
- Hurtado AM, Frey M, Hill K, Hurtado I, Baker J (2008) The role of helminthes in human evolution: Implications for global health in the 21st century. In: Elton S, O'Higgins P, eds. Medicine and evolution: current applications, future prospects Boca Raton, FL:Taylor and Francis Group. pp 153–180.
- Mutapi F, Ndhlovu PD, Hagan P, Woolhouse MEJ (1997) A comparison of humoral responses to *Schistosoma haematobium* in areas with low and high levels of infection. Parasite Immunol 19: 255–263. doi:10.1046/j.1365-3024.1997.d01-206.x.
- Needham CS, Lillywhite JE, Didier JM, Bianco AE, Bundy DAP (1993) Agedependency of serum isotype responses and antigen recognition in human whipworm (*Trichuris trichiura*) infection. Parasite Immunology 15: 683–692. doi:10.1111/j.1365-3024.1993.tb00583.x.
- Faulkner H, Turner J, Kamgno J, Pion SD, Boussinesq M, et al. (2002) Age- and infection intensity-dependent cytokine and antibody production in human trichuriasis: the importance of IgE. J Infect Dis 185: 665–72. doi:10.1086/ 339005.
- Ndhlovu P, Cadman H, Vennervald BJ, Christensen NO, Chidimu M, et al. (1996) Age-related antibody profiles in *Schistosoma haematobium* infections in a rural community in Zimbabwe. Parasite Immunol 18: 181–91.
- Hagan P, Blumenthal UJ, Dunn D, Simpson AJ, Wilkins HA (1991) Human IgE, IgG4 and resistance to reinfection with *Schistosoma haematobium*. Nature 349: 243–5. doi:10.1038/349243a0.
- Bundy D a, Lillywhite JE, Didier JM, Simmons I, Bianco a E (1991) Agedependency of infection status and serum antibody levels in human whipworm (*Trichuris trichiura*) infection. Parasite Immunol 13: 629–38.
- Cooper PJ, Alexander N, Moncayo A-L, Benitez SM, Chico ME, et al. (2008) Environmental determinants of total IgE among school children living in the rural Tropics: importance of geohelminth infections and effect of anthelmintic treatment. BMC Immunol 9: 33. doi:10.1186/1471-2172-9-33.
- Hagel I, Lynch NR, Prisco MC, Rojas E, Pérez M, et al. (2008) Ascaris reinfection of slum children: relation with the IgE response. Clin Exp Immunol 94: 80–83. doi:10.1111/j.1365-2249.1993.tb05981.x.

growth. In Shuar children high IgE levels are associated with increased stunting [18]. It seems plausible that insults to growth may be most pronounced in populations in which peak infection rates occur during critical growth periods, such as early adolescence. Additionally, the timing of infection may affect the development of immune function in other ways, for example by affecting the $T_H 1/T_H 2$ balance, with consequences for the later development of allergy [71]. Although these hypotheses remain to be tested, they suggest that interventions might be developed with the specific goal of shifting infection peaks toward less critical ages.

Acknowledgments

The authors thank the Federación Interprovincial de Comunidades Shuar (FICSH); Ministerio de Salud Publica, Area 6, Morona Santiago, Ecuador; Gran Consejo Tsimane; the members of the Tsimane Health and Life History Project, especially Dr. Daniel Eid, John Aguilar, and Ivan Maldonado; Collaborators and informants: Oswaldo Mankash; Cesar Kayap; Dr. Otto Campana; Medardo Tunki; Berta Fernandez and the Jempekat family; Tara Cepon; Tiffany Gandolfo; Julian Lim; and Heather Shattuck-Faegre. Finally, we thank two anonymous reviewers for their helpful comments.

Author Contributions

Conceived and designed the experiments: ADB MDG LSS HSK JJS. Performed the experiments: ADB MDG LSS FCM MAL MAM HSK JJS. Analyzed the data: ADB FCM MAL MAM. Contributed reagents/ materials/analysis tools: MDG LSS HSK JSS. Wrote the paper: ADB MDG JJS.

- Hagel I, Cabrera M, Sánchez P, Rodríguez P, Lattouf JJ (2006) Role of the low affinity IgE receptor (CD23) on the IgE response against Ascaris lumbricoides in Warao Amerindian children from Venezuela. Investigación Clínica 47: 241–51.
- Iancovici Kidon M, Stein M, Geller-Bernstein C, Weisman Z, Steinberg S, et al. (2005) Serum immunoglobulin E levels in Israeli-Ethiopian children: environment and genetics. Isr Med Assoc J 7: 799–802.
- Sheldon BC, Verhulst S (1996) Ecological immunology: costly parasite defences and trade-offs in evolutionary ecology. Trends Ecol Evol 11: 317–321. doi:10.1016/0169-5347(96)10039-2.
- Lochmiller RL, Deerenberg C (2000) Trade-offs in evolutionary immunology: just what is the cost of immunity? Oikos 88: 87–98. doi:10.1034/j.1600-0706.2000.880110.x.
- Blackwell AD, Snodgrass JJ, Madimenos FC, Sugiyama LS (2010) Life history, immune function, and intestinal helminths: Trade-offs among immunoglobulin E, C-reactive protein, and growth in an Amazonian population. Am J Hum Biol 22: 836–48. doi:10.1002/ajhb.21092.
- Elias D, Britton S, Aseffa A, Engers H, Akuffo H (2008) Poor immunogenicity of BCG in helminth infected population is associated with increased in vitro TGFbeta production. Vaccine 26: 3897–3902. doi:10.1016/j.vaccine.2008.04.083.
- Hurtado AM, Hurtado I, Hill K (2004) Public health and adaptive immunity among natives of South America. In: Salzano FM, Hurtado AM, eds. Lost paradises and the ethics of research and publication Oxford University Press: New York. pp 164–90.
- Riet E van, Adegnika AA, Retra K, Vieira R, Tielens AGM, et al. (2007) Cellular and humoral responses to influenza in gabonese children living in rural and semi-urban areas. Journal Infect Dis 196: 1671–8. doi:10.1086/522010.
- Hotez PJ, Brindley PJ, Bethony JM, King CH, Pearce EJ, et al. (2008) Helminth infections: the great neglected tropical diseases. J Clin Invest 118: 1311–1321. doi:10.1172/JCI34261.
- Johnson CC, Peterson EL, Ownby DR (1998) Gender differences in total and allergen-specific immunoglobulin E (IgE) concentrations in a population-based cohort from birth to age four years. Am J Epidemiol 147: 1145–52.
- Lindberg R, Arroyave C (1986) Levels of IgE in serum from normal children and allergic children as measured by an enzyme immunoassay. J Allergy Clin Immunol 78: 614–618. doi:10.1016/0091-6749(86)90078-3.
- Descola P (1996) The spears of twilight: life and death in the Amazon jungle. New Press, New York. 458 p.
- Harner MJ (1984) The Jivaro, people of the sacred waterfalls. University of California Press, Berkeley. xx, 233 p., [18] p. of plates p.
- Blackwell AD, Pryor III G, Pozo J, Tiwia W, Sugiyama LS (2009) Growth and market integration in Amazonia: A comparison of growth indicators between Shuar, Shiwiar, and nonindigenous school children. Am J Hum Biol 21: 161–171. doi:10.1002/ajhb.20838.

- 28. Sackey ME, Weigel MM, Armijos RX (2003) Predictors and nutritional consequences of intestinal parasitic infections in rural Ecuadorian children. J Trop Pediatrics 49: 17-23. doi:10.1093/tropej/49.1.17.
- 29. San Sebastian M, Santi S (2000) Control of intestinal helminths in schoolchildren in Low-Napo, Ecuador: impact of a two-year chemotherapy program. Revista da Sociedade Brasileira de Medicina Tropical 33: 69-73.
- 30. Gurven M, Kaplan H, Winking J, Eid Rodriguez D, Vasunilashorn S, et al. (2009) Inflammation and infection do not promote arterial aging and cardiovascular disease risk factors among lean horticulturalists. PLoS ONE 4: e6590. doi:10.1371/journal.pone.0006590.
- 31. Gurven M, Kaplan H, Winking J, Finch C, Crimmins EM (2008) Aging and inflammation in two epidemiological worlds. J Gerontol A Biol Sci Med Sci 2008 63 63: 196-199
- 32. McDade TW, Leonard WR, Burhop J, Reves-Garciá V, Vadez V, et al. (2005) Predictors of C-reactive protein in Tsimane' 2 to 15 year-olds in lowland Bolivia. Am J Phys Anthropol 128: 906-913. doi:10.1002/ajpa.20222.
- 33 Vasunilashorn S, Crimmins EM, Kim JK, Winking J, Gurven M, et al. (2010) Blood lipids, infection, and inflammatory markers in the Tsimane of Bolivia. Am J Hum Biol 22: 731-740. doi:10.1002/ajhb.21074.
- 34. Tanner S, Leonard WR, Mcdade TW, Reyes-Garcia V, Godoy R, et al. (2009) Influence of helminth infections on childhood nutritional status in lowland Bolivia. Am J Hum Biol 21: 651-656. doi:10.1002/ajhb.20944.
- 35. Benefice E, Monroy SL, Jiménez S, López R (2006) Nutritional status of Amerindian children from the Beni River (lowland Bolivia) as related to environmental, maternal and dietary factors. Public Health Nutrition 9: 327-335. doi:10.1079/PHN2005852.
- 36. Foster Z, Byron E, Reyes-García V, Huanca T, Vadez V, et al. (2005) Physical growth and nutritional status of Tsimane' Amerindian children of lowland Bolivia. Am J Phys Anthropol 126: 343-351. doi:10.1002/ajpa.20098
- 37. McDade TW, Williams S, Snodgrass JJ (2007) What a drop can do: dried blood spots as a minimally invasive method for integrating biomarkers into populationbased research. Demography 44: 899-925.
- 38. Tanner S, McDade TW (2007) Enzyme immunoassay for total immunoglobulin E in dried blood spots. Am J Hum Biol 19: 440-442. doi:10.1002/ajhb.20635.
- 39. Terhell AJ, Price R, Koot JWM, Abadi K, Yazdanbakhsh M (2001) The development of specific IgG4 and IgE in a paediatric population is influenced by filarial endemicity and gender. Parasitology 121: 535-543. doi:10.1017/ S0031182099006617
- 40. Stapel SO, Eysink PED, Vrieze J, Aalberse RC (2004) IgE testing in capillary blood. Pediatric Allergy Immunol 15: 230-233. doi:10.1111/j.1399-3038.2004.00142.x
- Eberl M, Al-Sherbiny M, Hagan P, Ljubojevic S, Thomas AW, et al. (2002) A 41. novel and sensitive method to monitor helminth infections by faecal sampling. Acta Tropica 83: 183-187. doi:10.1016/S0001-706X(02)00089-X.
- 42. Gurven M, Kaplan H, Supa AZ (2007) Mortality experience of Tsimane Amerindians of Bolivia: regional variation and temporal trends. Am J Hum Biol 19: 376-98. doi:10.1002/ajhb.20600.
- 43. Wood SN (2006) Generalized additive models: an introduction with R. CRC Press.
- 44. Hastie T, Tibshirani R (1986) Generalized additive models. Statistical Science 1: 297 - 310.
- 45. Wood SN (2008) Fast stable direct fitting and smoothness selection for generalized additive models. J Royal Statist Soc B 70: 495. doi:10.1111/ 1467-9868.2007.00646.x.
- Wood SN (2003) Thin Plate Regression Splines. J Royal Statist Soc B 65: 95-114. doi:10.1111/1467-9868.00374.
- 47. Bergmann RL, Schulz J, Gunther S, Dudenhausen JW, Bergmann KE, et al. (1995) Determinants of cord-blood IgE concentrations in 6401 German neonates. Allergy 50: 65-71. doi:10.1111/j.1398-9995.1995.tb02484.x.
- 48. Bjerke T, Hedegaard M, Henriksen TB, Nielsen BW, Schiotz PO (1994) Several genetic and environmental factors influence cord blood IgE concentration. Pediatric Allergy Immunol 5: 88-94. doi:10.1111/j.1399-3038.1994.tb00223.x.
- 49. Croner S, Kjellman NI, Eriksson B, Roth A (1982) IgE screening in 1701 newborn infants and the development of atopic disease during infancy. Br Med J 57: 364-368. doi:10.1136/adc.57.5.364

- 50. Holt PG, Jones CA (2000) The development of the immune system during pregnancy and early life. Allergy: Europ J Allergy Clin Immunol 55: 688-697. doi:10.1034/j.1398-9995.2000.00118.x.
- 51. Lopez N, Barros-Mazon S de, Marluce Dos Santos Vilela M, Condino Neto A, Ribeiro ID (2002) Are immunoglobulin E levels associated with early wheezing? A prospective study in Brazilian infants. Europ Respir J 20: 640. doi:10.1183/ 09031936.02.00219302.
- 52. Petridou E, Kanariou M, Liatsis M, Spanou K, Revinthi K, et al. (1995) Factors influencing serum immunoglobulin E levels in Greek children. Allergy 50: 210-214. doi:10.1111/j.1398-9995.1995.tb01135.x.
- 53. King CL, Malhotra I, Mungai P, Wamachi A, Kioko J, et al. (1998) B cell sensitization to helminthic infection develops in utero in humans. J Immunol 160: 3578-3584.
- Weil GJ, Hussain R, Kumaraswami V, Tripathy SP, Phillips KS, et al. (1983) 54. Prenatal allergic sensitization to helminth antigens in offspring of parasiteinfected mothers. J Clin Invest 71: 1124. doi:10.1172/JCI110862.
- Walter SD, Feinstein AR, Wells CK (1987) Coding ordinal independent 55. variables in multiple regression analyses. Am J Epidemiol 125: 319.
- Akaike H (1974) A new look at the statistical model identification. IEEE Trans 56. Automatic Control 19: 716-723.
- 57. Buckley CE, Larrick JW, Kaplan JE (1985) Population differences in cutaneous methacholine reactivity and circulating IgE concentrations. J Allergy Clin Immunol 76: 847 doi:10.1016/0091-6749(85)90759-6
- Kaplan JE, Larrick JW, Yost JA (1980) Hyperimmunoglobulinemia E in the 58. Waorani, an isolated Amerindian population. Am Journal Trop Med Hygiene 29: 1012-1017.
- 59. Kron MA, Ammunariz M, Pandey J, Guzman JR (2000) Hyperimmunoglobulinemia E in the absence of atopy and filarial infection: the Huaorani of Ecuador. Allergy Asthma Proc 21: 335–341. doi:10.2500/108854100778249060.
- 60. Grundbacher F (1975) Causes of variation in serum IgE levels in normal populations. J Allergy Clin Immunol 56: 104-111. doi:10.1016/0091-6749(75)90114-1.
- 61. Dodig S, Richter D, Benko B, Zivcic J, Raos M, et al. (2006) Cut-off values for total serum immunoglobulin E between non-atopic and atopic children in northwest Croatia. Clin Chem Laborat Med 44: 639
- 62. Woolhouse MEJ (1992) A theoretical framework for the immunoepidemiology of helminth infection. Parasite Immunology 14: 563-578. doi:10.1111/j.1365-3024.1992.tb00029.x.
- 63. Warren K (1974) Helminthic diseases endemic in the United States. Am J Trop Med Hyg 23: 723-730.
- 64. Hotez PJ (2008) Neglected infections of poverty in the United States of America. PLoS Neglected Tropical Diseases 2: e256. doi:10.1371/journal.pntd.0000256.
- 65. Kappus KD, Lundgren RG, Juranek DD, Roberts JM, Spencer HC (1994) Intestinal Parasitism in the United States: Update on a Continuing Problem. Am J Trop Med Hyg 50: 705-713.
- Amin OM (2002) Seasonal prevalence of intestinal parasites in the United States 66. during 2000. Am J Trop Med Hyg 66: 799-803.
- 67. McSharry C, Xia Y, Holland CV, Kennedy MW (1999) Natural immunity to Ascaris lumbricoides associated with immunoglobulin E antibody to ABA-1 allergen and inflammation indicators in children. Infect Immun 67: 484-9.
- 68. Hagel I, Cabrera M, Buvat E, Gutiérrez L, Santaella C, et al. (2008) Antibody responses and resistance against Ascaris lumbricoides infection among Venezuelan rural children: the influence of ethnicity. J Trop Pediat 54: 354-6. doi:10.1093/tropej/fmn032.
- Cooper PJ, Ayre G, Martin C, Rizzo JA, Ponte EV, et al. (2008) Geohelminth 69. infections: a review of the role of IgE and assessment of potential risks of anti-IgE treatment. Allergy 63: 409-17. doi:10.1111/j.1398-9995.2007.01601.x.
- Perlmann H, Helmby H, Hagstedt M, Carlson J, Larsson PH, et al. (1994) IgE elevation and IgE anti-malarial antibodies in Plasmodium falciparum malaria: association of high IgE levels with cerebral malaria. Clin Exper Immunol 97: 284-292. doi:10.1111/j.1365-2249.1994.tb06082.x.
- 71. Yazdanbakhsh M, Wahyuni S (2005) The role of helminth infections in protection from atopic disorders. Cur Opin Allergy Clin Immunol 5: 386-391.